Author's response to reviews

Title: The prevalence of injection-site reactions with disease-modifying therapies and their effect on adherence in patients with multiple sclerosis: an observational study

Authors:

Karsten Beer (karstenbeer@hotmail.com)
Martin Müller (martin.mueller@ksl.ch)
Anna Marie Hew-Winzeler (a.m.hew@bluewin.ch)
Adriano Bont (abont@hin.ch)
Philippe Maire (philippe.maire@hin.ch)
Xiaojun You (Phil.You@Biogenidec.com)
Pamela Foulds (Pamela.Foulds@Biogenidec.com)
Jessica Mårland (jmarlind@biogen-dompe.ch)
Daniela Curtius (dcurtius@biogen-dompe.ch)

Version: 5 Date: 7 November 2011

Author's response to reviews:

The presentation of the P values in Figure 1 has been made consistent with the text. The y-axis labels in Figure 2 have also been clarified. In addition, the parenthetical/bracketed instance of "chi-square test" in the sentence in the text referring to Figure 4 has been put after the P value, as it appears elsewhere.